Author's response to reviews

Title: GenexGene interaction between MnSOD and GPX-1 and breast cancer risk: a nested case-control study

Authors:

David G Cox (dcox@hsph.harvard.edu)
Rulla M Tamimi (nhrmt@channing.harvard.edu)
David J Hunter (david.hunter@channing.harvard.edu)

Version: 4 Date: 28 August 2006

Author's response to reviews:

The changes requested by the editorial office were completed. The conclusion section was added to the abstract. IRB approval information has been added to the methods. Proper section headings have been added.